Author's response to reviews

Title: Variables with time-varying effects and the Cox model: Some statistical concepts illustrated with a prognostic factor study in breast cancer

Authors:

Carine A Bellera (bellera@bergonie.org)
Gaëtan MacGrogan (macgrogan@bergonie.org)
Marc Debled (debled@bergonie.org)
Christine Tunon de Lara (tunon@bergonie.org)
Véronique Brouste (brouste@bergonie.org)
Simone Mathoulin-Pélissier (mathoulin@bergonie.org)

Version: 7 Date: 12 March 2010

Author's response to reviews:

The title 'WORKING EXAMPLE' was removed in several occasions. We have thus put back these 2 words in several places. Please, do not remove them.

There were also some typos that we corrected (an 's' had been forgotten, two commas, etc).